Author’s response to reviews

Title: PROtective Ventilation with a low versus high Inspiratory Oxygen fraction (PROVIO) and its effects on postoperative pulmonary complications: protocol for a randomized controlled trial

Authors:
Xue-Fei Li (kikyou015@163.com)
Dan Jiang (631688232@qq.com)
Yu-Lian Jiang (liano5@foxmail.com)
Hong Yu (happyjia1990@foxmail.com)
Jia-Li Jiang (834143383@qq.com)
Lei-Lei He (171764169@qq.com)
Xiao-Yun Yang (cdyangxiaoyun@163.com)
Hai Yu (yuhaishan117@yahoo.com)

Version: 3 Date: 04 Aug 2019

Author’s response to reviews:

Responses to the reviewers’ comments
We would appreciate the editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript. Our point-by-point responses to the reviewers’ comments are as follows:

Reviewer #2:
1. On page 9 under the heading "Intraoperative care", the "v" of vasoactive drugs is written with a smaller font (10.5 times new roman), please write it in the same font as the manuscript.
Response: Thanks for your correction. We have revised the word (page 9, line 28).

Additional change:
We have checked the references.

We thank you very much for giving us an opportunity to revise our manuscript. We have studied reviewers’ comments carefully and tried our best to have made revision which marked in red in the paper.